Author's response to reviews

Title: Supervised and non-supervised Nordic Walking in the treatment of chronic low back pain A single blind randomized clinical trial.

Authors:

Jan Hartvigsen (jhartvigsen@health.sdu.dk)
Lars Morsoe (lmorsoe@shf.regionsyddanmark.dk)
Tom Bendix (tbendix@health.sdu.dk)
Claus Manniche (claus.manniche@shf.regionsyddanmark.dk)

Version: 4 Date: 11 January 2010

Author's response to reviews:

Dear Dr Graham,

Thank you for your kind letter and for accepting our paper for publication in BMC Musculoskeletal Disorders. In accordance with the remarks by the Associate Editor we have revised and added the following to the paper:

1. The reference list is revised and formatted and should now be correct. In addition we have accessed the URL in reference one and updated it.

2. We have clarified the flow of participants in the text and corrected the numbers in the flow chart (Figure 1).

3. We have changed the legend to Figure 5 and clarified in the text on page 14. The figure represents the range of activity in the supervised and non-supervised groups by plotting the least and most active participants in the two groups over a day. The average intensity is reported in the text. The terms in the figure proper have also been changed accordingly.

4. We have gone through the manuscript formatting checklist and formatted the manuscript where necessary.

We trust that the paper is now acceptable. If not, please don’t hesitate to get back to us.

With best regards,
Jan Hartvigsen